Author's response to reviews

Title: Economic Impacts of Illness in Older Workers: Quantifying the impact of illness on income, tax revenue and government spending

Authors:

Deborah J Schofield (deborah.schofield@ctc.usyd.edu.au)
Rupendra N Shrestha (rupendra.shrestha@ctc.usyd.edu.au)
Richard Percival (richard.percival@natsem.canberra.edu.au)
Megan E Passey (megan.passey@ucrh.edu.au)
Simon J Kelly (simon@kellyresearch.com.au)
Emily J Callander (emily.callander@ctc.usyd.edu.au)

Version: 4 Date: 30 May 2011

Author's response to reviews:

Formatting changes to be made:

Major revisions (we require the author to make these changes):

References: the web link in reference 11 is still not working. Please provide a working link.

- The link is correct, and is working correctly -

Authors' contributions: Please include the sentence "All authors read and approved the final manuscript" in this section.

- this has been added

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above):

Figures: Please crop figure 1 so there is no white space around the figure.

- this has been done

Blank page: please remove blank page 21.
- this has been removed